Mauna Loa Solar Observatory Observer's Log ______ Wed Apr 11 17:32:07 GMT 2018 Year: 18 Doy: 101 Observer: berkey WEATHER COMMENT: berkey: Wed Apr 11 17:32:58 GMT 2018 Temp: 42.5f, Humidity: 76%, Pressure: 28.826in, Wind: 10mph from 137degs, Skies: clear end Wed Apr 11 17:52:48 GMT 2018 Kcor Focus/alignment program exited Wed Apr 11 17:56:24 GMT 2018 KCOR Start Synoptic Patrol Wed Apr 11 18:26:48 GMT 2018 KCOR End Patrol Wed Apr 11 23:41:38 GMT 2018 Kcor Focus/alignment program exited Thu Apr 12 00:38:50 GMT 2018 Kcor Focus/alignment program exited Thu Apr 12 00:43:48 GMT 2018 SGS Alignment complete Thu Apr 12 00:43:49 GMT 2018 SGS Alignment complete Thu Apr 12 00:43:50 GMT 2018 Kcor Focus/alignment program exited GENERAL COMMENT BY berkey: Thu Apr 12 02:10:03 GMT 2018 Not much data today but we got a couple kcor images between clouds. Some clean up/recovery for the CoMP decomissioning today tables returned to NOAA and webcams retrived from the dome. end

GENERAL COMMENT BY berkey: Thu Apr 12 02:13:54 GMT 2018 Spar cam was reinstalled this morning on the Chromag face of the spar. It may be a bit lower on the spar.

end ONSITE STAFF: berkey